Author's response to reviews

Title: The Nakuru Eye Disease Cohort Study: Methodology & Rationale

Authors:

Andrew Bastawrous (andrew.bastawrous@lshtm.ac.uk)
Wanjiku Mathenge (ciku@email.com)
Tunde Peto (tunde.peto@moorfields.nhs.uk)
Helen A Weiss (Helen.weiss@lshtm.ac.uk)
Hillary K Rono (hkrono75@gmail.com)
Allen Foster (Allen.Foster@lshtm.ac.uk)
Matthew J Burton (Matthew.Burton@lshtm.ac.uk)
Hannah Kuper (Hannah.Kuper@lshtm.ac.uk)

Version: 2
Date: 13 March 2014

Author's response to reviews:

1. Corresponding author indicated
2. Abstract reformatted
3. Figures removed and uploaded separately
4. Tables removed and uploaded separately
5. Abbreviations moved
6. Appendix added
7. Font changed to Ariel
8. Ethics approval letters emailed